Author's response to reviews

Title: Cefditoren: Comparative efficacy with other antimicrobials and risk factors for resistance in clinical isolates causing UTIs in outpatients

Authors:

Despina Hatzaki (aml@otenet.gr)
Garyphallia Poulakou (gpoulakou@gmail.com)
Ioannis Katsarolis (ikatsarolis@gmail.com)
Niki Lambri (lampriniki@yahoo.gr)
Maria Souli (Msouli@med.uoa.gr)
Ioannis Deliolanis (idediol@gmail.com)
Georgios K Nikolopoulos (gknikolopoulos@gmail.com)
Evangelia Lebessi (elebesi@otenet.gr)
Helen Giamarellou (e.giamarellou@hygeia.gr)

Version: 2 Date: 25 March 2012

Author's response to reviews:

Dear Mr Editor,

On behalf of my co-authors I would like to submit for consideration the manuscript entitled: “Cefditoren: Comparative efficacy with other antimicrobials and risk factors for resistance in clinical isolates causing UTIs in outpatients”.

This is the revised version of the manuscript, according to the editorial requests. An addition explaining approval by Ethics Committee of participating centers was incorporated in section : 4.1. Study period and participating institutions.

The authors have seen and agreed to the revised version of the paper.

We remain at your disposition for any further clarifications.

Despina Hatzaki1,
Garyphallia Poulakou2, (corresponding author)
Ioannis Katsarolis2, Niki Lambri2,
Maria Souli2, Ioannis Deliolanis3,
Georgios K Nikolopoulos4, Evangelia Lebessi1,
Helen Giamarellou5, 6

(1) Department of Microbiology, “P. and A. Kyriakou” Children’s Hospital, (2) 4th Department of Internal Medicine and Infectious Diseases Research Laboratory, Athens University School of Medicine (3) Department of Microbiology, “Laikon” General Hospital of Athens (4) Hellenic Center for the Disease Control and Prevention (KEELPNO), (5) Athens University School of Medicine, (6) 6th Department of Internal Medicine, Hygeia Hospital, Athens, Greece